%File content.tex, 9 September, 1992
%\magnification=\magstep1
\input ij.sty
\input mssymb
\def\ {\thinspace}
\lineskiplimit=100pt 
\lineskip=7pt %This controls the spacing between indexitems of
%both kinds
\tenpoint
\null\vskip 3cm

\def\index\indexvolume#1\indexyear#2\endindextitle{
\rm\parindent=0pt%\overfullrule=0pt
\centerline{ISRAEL JOURNAL OF MATHEMATICS}
\vskip0.4cm
\centerline{TABLE OF CONTENTS, VOLUME #1\unskip, #2\unskip}
\vskip1.2cm
\frenchspacing}
\def\endindex{}

\def\indexitem\indexauthor#1\indextitle#2\indexpage#3\endindexitem
{\halign {\normalbaselines%
\vbox{\hsize=11cm\hangindent=2em\hangafter=1 ##}%
&\hbox to 1.3cm{\hfill ##}\cr%
{\smc #1\unskip}\quad{\sl #2} & #3 \cr}\par}
\def\indexauthor{}
\def\endindexitem{}
\def\indexpage{}
\def\indextitle{}

\def\indexsee\indexauthor#1\partner#2\endindexitem
{{\smc #1\unskip\enspace}{\sl See }{\smc #2\unskip}\par}

\index
\indexvolume 112
\indexyear 1999
\endindextitle

\indexitem 
\indexauthor An, J.
\indextitle The Alperin and Dade conjectures for the simple Conway's
third group
\indexpage 109
\endindexitem

\indexitem
\indexauthor Buzzi, J.
\indextitle Markov extensions for multi-dimensional dynamical
systems
\indexpage 357
\endindexitem

\indexitem
\indexauthor Cohen, P.~B. and A.~Regev
\indextitle Asymptotics of multinomial sums and identities between
multi-integrals
\indexpage 301
\endindexitem

\indexitem
\indexauthor El Abdalaoui, E.~H.
\indextitle La singularit\'e mutuelle presque s\^ure du spectre des
transformations d'Ornstein
\indexpage 135
\endindexitem

\indexitem
\indexauthor Enriquez, B. and V.~Rubtsov
\indextitle Quasi-Hopf algebras associated with {\frak\$sl}$_2$ and
complex curves
\indexpage 61
\endindexitem

\indexitem
\indexauthor Hoffman, C.
\indextitle A loosely Bernoulli counterexample machine
\indexpage  237
\endindexitem 

\indexitem
\indexauthor Hoffman, C.
\indextitle A Markov random field which is $K$ but not Bernoulli
\indexpage 249
\endindexitem

\indexitem
\indexauthor Keen, L., J.~R.~Parker and C.~Series
\indextitle Combinatorics of simple\break closed curves on the twice
punctured torus
\indexpage 29
\endindexitem

\indexitem
\indexauthor Lapid, E.~M.
\indextitle Some results on multiplicities for SL($n$)
\indexpage 157
\endindexitem

\indexitem
\indexauthor Matou\v{s}ek, J. and E.~Matou\v{s}kov\'a
\indextitle A highly non-smooth norm on\break Hilbert space
\indexpage 1
\endindexitem

\indexsee
\indexauthor Matou\v{s}kov\'a, E.
\partner J.~Matou\v{s}ek
\endindexitem

\indexitem
\indexauthor Mekler, A., A.~Ros{\l}anowski and S.~Shelah
\indextitle On the $p$-rank of Ext
\indexpage 327
\endindexitem

\indexitem 
\indexauthor Montgomery, S. and H.-J.~Schneider
\indextitle Prime ideals in Hopf Galois extensions
\indexpage 187
\endindexitem

\indexitem 
\indexauthor Newelski, L.
\indextitle Vaught's conjecture for some meager groups
\indexpage 271
\endindexitem

\indexsee
\indexauthor Parker, J.~R.
\partner L.~Keen
\endindexitem

\indexsee
\indexauthor Regev, A.
\partner P.~B.~Cohen
\endindexitem

\indexsee
\indexauthor Ros{\l}anowski, A.
\partner A.~Mekler
\endindexitem

\indexsee
\indexauthor Rubtsov, V.
\partner B.~Enriques
\endindexitem

\indexsee
\indexauthor Schneider, H.-J.
\partner S.~Montgomery
\endindexitem

\indexsee
\indexauthor Series, C.
\partner L.~Keen
\endindexitem

\indexsee
\indexauthor Shelah, S.
\partner A.~Mekler
\endindexitem

\endindex
%End of file, content.tex
